Tutorial for week 4 (Jan 28-Feb 1, 2002)

January 29, 2002

The following questions will be considered during the tutorial sessions on Wednesday...

T4Q1: Given the frame .frm containing a canvas and a quit button, give sensible names for the canvas and the button.

T4Q2: Modify SimpleProg.tcl to have an extra button clear above the quit button which clears the date display.

T4Q3: Modify SimpleProg.tcl to have an extra button clear to the left of the quit button which clears the date display.

T4Q4: What is the effect of the following tcl command? set a [exec ls] T4Q5: What is the effect of the following tcl command? set a expr 3 + 4 T4Q6: Write a minimal Tk application which puts up a single File menu with a Quit item in it.